Author's response to reviews

Title: Wearable defibrillator use In heart Failure (WIF): Results of a prospective registry

Authors:

Andrew C Kao (akao@cc-pc.com)
Steven W Krause (stevekrause4601@yahoo.com)
Rajiv Handa (rhanda@heartspecialty.com)
Darshak Karia (darshakkaria@yahoo.com)
Guillermo Reyes (guillermoreyesmd@yahoo.com)
Nicole R Bianco (nbianco@zoll.com)
Steven J Szymkiewicz (sszymkiewicz@zoll.com)

Version: 3 Date: 24 July 2012

Author's response to reviews:

Dear (Miss) Marinette Lacson,

In response to your question "please can you revise your manuscript and add the name of ethics committee", we have made the following change to the methods section. As there were several IRBs that approved the study, we felt that this was the best way to explain it.

"The protocol, informed consent document, and relevant supporting information was submitted and approved for each study site by their local or a national institutional review board before the study was initiated in accordance with ethical principles that have their origin in the Declaration of Helsinki and consistent with Good Clinical Practice and applicable regulatory requirements."

Sincerely,
Nicole Bianco on behalf of Dr. Andrew Kao